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UK-V Cluster Meeting at RAL – Wed 8 Sep 2004
1. Preliminaries

1.1 Present

Glasgow

Craig Buttar, Sasha Tcheplakov

Lancaster

Alex Chilingarov, Ian Mercer, Peter Ratoff

Liverpool

Phil Allport, Neil Jackson

Manchester

Mike Ibbotson
RAL

Jeff Bizzell, Stephen Haywood, Jason Tarrant, Ian Wilmut

Sheffield

Richard French, Dan Tovey

Valencia

Carlos Lacasta

1.2 Apologies

Richard Bates, Ian Duerdoth, Jo Pater, Steve Snow, Tim Jones

1.3 Minutes of Sheffield Meeting (16 June 04)

To “Actions from the previous meeting” the following sentence was added:

Following recommendations of the ATLAS-UK Management Board the work on the X-ray survey at Lancaster is put on hold.

The minutes were accepted after minor corrections.

1.4 Actions from Sheffield Meeting

· Decide on whether UK-V makes CiS based modules.
Phil says “No”.
· Decide on whether to continue with outers or swap to middles.
Can swap, if necessary.
· Put trained detectors into disks for tests in cold room to investigate stability.
Discs will be tested anyway.
· How do we ensure this is entered in the DB ?
No way to ensure this, but people must do this.
· Sort-out how to handle re-work modules in DB
Joe has to reconstruct list from emails – issue being addressed.
· Where should 8C go, store at RAL ?
Now at Liverpool.
· Need tooling to remove modules before they are put on
Addressed and proved to the satisfaction of Review Committee.
1.5 Further Matters Arising from Sheffield Meeting
Stephen explained that the Disc Delivery Schedule has slipped by ~3 weeks and will be presented by Ian later.
2. Presentations

All presentations can be found at:

http://hepwww.rl.ac.uk/atlas-sct/engineering/meetings/uk-v_040908/web.htm
2.1 Overview – Phil Allport

While there has been talk of reshuffling of responsibilities, these have not been agreed by the funding agencies. Therefore, for the time being we are sticking with the agreed production numbers (shown by Phil). The status will be reviewed in October.

The agreed numbers contain a 20% contingency (15% yield and 5% mounting) for the Module starts. Jeff needs to identify 920 good Hybrids which will result in 855 Module starts. UK-V has produce 50% of the required number of starts, yielding 33% of the required “good” Modules.
More info is available on the production summary website:

http://ific.uv.es/~lacasta/ECproduction/
Yields have dropped slightly from the last meeting because Joe is now including ones before they even get to the test stage; some of these may be re-workable. UK-V is only achieving 85% yield if one includes “pass” Modules. Geneva have a very high yield since they count lost Hybrids and not 4-wafer assemblies. Their way of doing assembly wastes fewer good chips.

Production is still very sensitive to the supply of components. Spines are still holding things back.

R1) Glasgow: Check numbers going onto Joe’s web page.
There have been problems with noise associated with the Redundancy Links. This seems somewhat better with the 5 k resistor replaced with 0 . Ashley will investigate the LV0 to see whether that introduces extra noise.
The understanding of the contamination/erosion of tracks on one fan-in still unclear; a chemical analysis will be undertaken at CERN.
Prague CTU and Melbourne are still to qualify – Melbourne is being held back by Gareth’s presence at the Test-beam. The absence of CiS Modules from Melbourne may be compensated if the yield of Hamamatsu Modules increases. Production from each of the other clusters is now increasing. 
2.2 Components

2.2.1 Hybrids – Jeff Bizzell

Of the 920 Hybrids required, 510 have been shipped. Jeff can test 24/week; however the shipping rate cannot exceed this number due to the need to retest some Hybrids and the output is subject the supply of Hybrids from Cicorel. The number of returns has greatly reduced. Currently Jeff has enough to keep going, with a few being retested. He still expects to finish by the end of the year.
Some time towards end of year, we need to think whether RAL needs to take on further Hybrids.

2.2.2 Spines – Richard Bates 

No news.
2.2.3 Fan-ins – Miguel & Manuel, reported by Carlos
Currently 80% of the total number of Spines has been produced, and the complete production should be completed by end of year. Since the last SCT Week, the shipment plan has been adjusted to match the progress at the various sites.
2.3 Modules

2.3.1 Valencia – Carlos Lacasta
Valencia expect to increase their Module production rate to the nominal 6/week.

2.3.2 Manchester – Steve Snow, reported by Mike Ibotson
They are close to producing 240 Outers, however Phil pointed out the required target is 250. Production of Middles will start soon, with Inners starting in November. Spines are the most critical component, although Manchester have some in stock. There have been a small number of problems meeting the x-y specs – these problems have reduced significantly with time (apart from a couple of recent failures). The z metrology looks very good. 
2.3.3 Glasgow – Sasha Tcheplakov
2.3.4 Liverpool – Ashley Greenall
The status is essentially contained in Phil’s presentation.
2.4 Module QA – Joe Foster; presented by Mike Ibbotson
The key message is that people must keep Dbase up to date.
R2) All: Update Dbase as appropriate.
2.5 Production Schedule – Carlos Lacasta
Most clusters will finish by the end of May 2005; however the last Modules for Disc 1A will not be available until August (WAS THIS CORRECT?). Even if MPI and Freiberg reach their target production rate, they will not finish until June. By taking on extra production, Valencia & Nikhef could help to obtain completion by the end of May; although Nikhef want to finish by end of 2004 due to Disc/EC commitments.
Things may be sped up if we can utilise LV & HV cards from Barrel as they finish with them. Shaun Roe is investigating this possibility.
Phil pointed out that so far, in eth UK, we have no coherent plan for rework (mainly chip-changes).
2.6 Discs
2.6.1 Discs Assembly – Ian Wilmut
Discs 9C and 8C are now in Liverpool. RAL hope that 7C should be completed by around 17 September. Disc 4C has started today. RAL is 3-4 weeks late – while the time estimates for the assembly are reasonably good, and perhaps even conservative, the testing and precision placement is taking longer than foreseen, despite 2 weeks having been foreseen as a contingency.
The observation is that 9A is going slowly at Nikhef. Phil would like to understand the (realistic) expected rate of Discs at Nikhef to optimise the production and allocation of various Module types.
Cooling Circuit production was on target, but has been stalled for ~2 weeks due to the manufacturer failing to deliver CC Blocks.
2.6.2 Modules Mounting – Neil Jackson
At Liverpool, Disc 9 is being tested in its test frame, while the Cylinder is on the Assembly Frame. All quadrants have been cooled to -25oC.
There is an unpleasant distortion in the noise-occupancy plot, symptomatic of feedback at a low threshold (but well below the actual operating threshold, so not a problem).

Liverpool have Modules coming out of their ears (not literally). The critical ones will be Short-Middles for Disc 8 – people should not send more until these have arrived. Propose to use “weaker” Modules (up to 10 bad channels in 1500) for Disc 9 – however these ARE good modules. The reasoning is that at the highest ||, the tracking will be poorest, and being the first Modules handled, they are more liable to damage. 
2.7 Macro-Assembly 

2.7.1 End-cap Engineering – Jason Tarrant
Support Structures
The Contract is completed; the first Support Cylinder is at Liverpool with the Cu G&S Foil (the second is at Nikhef).
Services
The design of the Cooling Circuits (on Cylinder) is basically completed; and preparations are being made to bend the pipes. The lengths of the LMT’s and Fibre-Tapes have been checked and are in pre-production.
Thermal Enclosure
This is going into final design.
The spare capillaries from Discs 9, 8 & 7 will be used to cool the LMT’s. Peter Ford will investigate perforations in the foil wrapping to avoid moisture retention.

PPF1
Trays and holder for LMT’s have been produced (just need to be tidied up) – this has some flexibility for variations in length of LMT’s. Lancaster are making temporary PPF1’s; these will be replaced at the time of the Integration in the SR Building in CERN.
2.7.2 Disc Insertion – Richard French
The Thermal End-plate has been finished at Sheffield. The Tooling is now complete and Disc Insertion has been proved. At the first attempt, an alignment of the Discs to 0.5 mm was achieved; with more fine tuning, 0.1 mm should be achievable.
Tests to do:

· 2nd test of Disc Grabber – end of September.
· Mounting/demounting of beam – should be done at Nikhef; but will also test at Liverpool
· Z-alignment tool from Nikhef.

For Services:
· Understand DCS

· Design Fibre Holder

· Understand Cooling connections etc

· Need services to test placement

If there is a quiet period the Cooling Interconnects could be added, otherwise this will have to be done at CERN 
An electrical string-test from the Hybrid to the TPPC has been carried out.
Craig hopes to insert a populate Disc at the end of October. There is concern that there should be suitable protection to avoid dropping parts into Cylinder – some thought is being given to this.
R3) Craig: Produce a Schedule which indicates when everything is ready/needed. 
This should help identify when key components are needed and should go as far as the insertion and read-out of a Disc
2.8 Preparation at CERN – Jo Pater
System-test sector

· As all working RODs are at H8, Jo is now working on getting the old mustard/optif readout going again. 

· Also she is working on getting PS0 to work with it; having some problems with that right at the moment.

SR1

· The Thermal Enclosure is finished except to cut holes for passage of power cables. 

· Enough cables for any single barrel are laid and tested and ready for the Barrel arrival. 

· Fibres are laid and tested but need some slight adjustments in order to reach the correct ROD crates. Canbus connections for 44 power supply crates ready and waiting. 

· Still waiting for the cooling.

3. Conclusions

3.1 AOB

Lancaster would like a decision about the X-ray survey by the end of October. Lancaster have not made progress because of the doubts. If the X-ray equipment is to be available for the End-cap, there needs to be an immediate green light.
R4) Peter: Email SG (with copies to Ian, Neil and Janet) asking for a prompt decision.

R5) Stephen: Remind Mike & Richard Nickerson of the benefits of X-ray survey.
3.2 Next Meeting

In 2 months in Liverpool, with the exact date to be decided.
R6) Ian: Clarify date of next meeting.

It would be good to visit Valencia soonish, perhaps in the winter.
3.3 Summary of Actions

R1) Glasgow: Check numbers going onto Joe’s web page.

R2) All: Update Dbase as appropriate.

R3) Craig: Produce a Schedule which indicates when everything is ready/needed. 
R4) Peter: Email SG (with copies to Ian, Neil and Janet) asking for a prompt decision.

R5) Stephen: Remind Mike & Richard Nickerson of the benefits of X-ray survey.

R6) Ian: Clarify date of next meeting.
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